Author's response to reviews

Title: Can the FAST and ROSIER adult stroke recognition tools be applied to confirmed childhood arterial ischemic stroke?

Authors:

Adriana Yock-Corrales (adriyock@gmail.com)
Franz E Babl (Franz.Babl@rch.org.au)
Ian T Mosley (ian.mosley@monash.edu)
Mark T Mackay (Mark.MacKay@rch.org.au)

Version: 4 Date: 19 October 2011

Author's response to reviews:

Formatting changes to be made:

Abstract: This should be identical in the manuscript and the submission system. Currently they are not. DONE

Structure: Change contributor statement to authors contributions. DONE
Change bibliography to references. DONE

References: the reference list should contain up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors. DONE

Tables: Please remove vertical lines. DONE